Author’s response to reviews

Title: Accuracy of endoscopic ultrasound-guided fine-needle aspiration in the suspicion of pancreatic metastases.

Authors:

José C Ardengh (jcelso@uol.com.br)
César V Lopes (drcvlopes@gmail.com)
Rafael Kemp (rafaelkemp@terra.com.br)
Filadélfio Venco (filadelfio.venco@diagnostika.med.br)
Eder R de Lima-Filho (ederrios1@yahoo.com.br)
José S dos Santos (jsdosanto@fmrp.usp.br)

Version: 3 Date: 9 March 2013

Author’s response to reviews:

We are sending to BMC Gastroenterology the paper entitled “Micro-histology of pancreatic masses submitted to endoscopic ultrasound-guided fine-needle aspiration in the suspicion of metastatic disease.” wrote by José Celso Ardengh, César Vivian Lopes, Rafael Kemp, Filadélfio Venco, Eder Rios de Lima-Filho, and José Sebastião dos Santos. All the authors colaborated equally for the elaboration of the manuscript, and nothing of this material has been published or is under evaluation for publication in other scientific journal. Besides, this study did not receive any grants or financial support. We believe this is going to be a very interesting reading for the BMC Gastroenterology readers, demonstrating the role of the endoscopic ultrasound-guided fine needle aspiration in the presence of pancreatic masses suspicious for metastatic disease in patients with history of a previous extrapancreatic cancer. We would like to suggest the following names for potential peer reviewers for our manuscript: 1) Todd Baron (Baron.Todd@mayo.edu); 2) Marcel Autran Machado (dr@drmarcel.com.br); 3) Jesse Lachter (ramila@netvision.net.il)